\begin{table}
\caption{\label{tab:Cstars_heavy}Abundances of neutron-capture elements in carbon enhanced metal-poor stars from literature. Columns (17), (19), (21), (23), (25) list random errors (when available). (27) Solar abundances reference adopted in the original reference source: \citet[][An89]{isotop_frac}, \citet[][Gr96]{Grevesse1996}, \citet[][Gr98]{Grevesse1998}, \citet[][Gr20]{Grevesse2000}, \citet[][As05]{Asplund2005}. references: (1) \citet{Cohen2004}, (2) \citet{Cohen2006}, (3) \citet{Lucatello2003}, (4) \citet{Aoki2007}, (5) \citet{Barklem2005}, (6) \citet{LucatelloPhD}, (7) \citet{Christlieb2002}, (8) \citet{Christlieb2004}, (9) \citet{Jonsell2006}, (10) \citet{Norris2007}, (11) \citet{Cohen2008}, (12) \citet{Goswami2006}, (13) \citet{Frebel2007}, (14) \citet{Frebel2005}, (15) \citet{Aoki2006}, (16) \citet{Frebel2006}, (17) \citet{Cohen2003}, (18) \citet{Johnson2002}, (19) \citet{TsangaridesPhD}, (20) \citet{Aoki2002c}, (21) \citet{Aoki2002b}, (22) \citet{Aoki2002d}, (23) \citet{Preston2001}, (24) \citet{Sneden2003_CS22892}, (25) \citet{Cayrel2004}, (26) \citet{Spite2005}, (27) \citet{Spite2006}, (28) \citet{Barbuy2005}, (29) \citet{Francois2007}, (30) \citet{Depagne2002}, (31) \citet{Sneden2003}, (32) \citet{Norris1997b}, (33) \citet{Bonifacio1998}, (34) \citet{Aoki2002a}, (35) \citet{Sivarani2006}, (36) \citet{Sivarani2004}, (37) \citet{Ivans2005}, (38) \citet{Aoki2004}, (39) \citet{Johnson2004}, (40) \citet{Zacs1998}, (41) \citet{Aoki2001}, (43) \citet{Plez2005}, (44) \citet{Deroo2005}, (45) \citet{Honda2004}, (46) \citet{Allen2006a}, (47) \citet{Christlieb2004}, (48) \citet{Hill2002}, (49) \citet{Plez2004}, (50) \citet{Honda2006}, (51) \citet{Roederer2008}, (52) Paper~II.}
\begin{tabular}{ l c c c c c c c c c c c c l}
\hline\hline
Object              & [Ba/Fe]       &  $\sigma$ &  [La/Fe]  &  $\sigma$ & [Ce/Fe] & $\sigma$ &   [Eu/Fe]   & $\sigma$ &  [Pb/Fe]   & $\sigma$ & Class &    Solar ref &  References        \\
&  (16)         &  (17)     &  (18)     &  (19)     &   (20)  &  (21)    &  (22)       &  (23)    & (24)       &  (25)    & (26) &  (27)  & (28)      \\
\hline 
  HE 0007-1832      &          0.23 &   ...  &    $<$1.80 &   ...  &     ...  &   ...  &       $<$1.86 &   ...  &    $<$3.21 &   ...  &    no &   An89 & (1)                                                                                                                   \\                                                                                            
  HE 0012-1441      &          1.22 &   0.27 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &    $<$2.09 &   ...  &     s &   An89 & (2)                                  							       			     \\
  HE 0024-2523      &          1.52 &   0.20 &         1.77 &   0.20 &     ...  &   ...  &       $<$1.07 &   0.10 &         3.32 &   0.10 &     s &   An89 & (3)                                  							       			     \\
  HE 0107-5240      &     $<$0.82 &   ...  &         ...  &   ...  &     ...  &   ...  &       $<$2.80 &   ...  &         ...  &   ...  &    no &   Gr98 & (7,8)              							       			     \\
  HE 0131-3953      &          2.20 &   0.23 &         1.94 &   0.27 &     1.93 &   0.29 &            1.62 &   0.28 &         ...  &   ...  &    rs &   Gr98 & (5)                                  							       			     \\
  HE 0143-0441      &          2.39 &   0.12 &         1.77 &   0.15 &     2.00 &   0.21 &            1.53 &   0.17 &         3.28 &   ...  &    rs &   An89 & (2)                                  							       			     \\
                    &          2.38 &   0.18 &         1.96 &   0.14 &     2.20 &   0.21 &            1.72 &   0.17 &         3.67 &   ...  &    rs &   An89 & (1)                                  							       			     \\
  HE 0202-2204      &          1.41 &   0.22 &         1.36 &   0.34 &     1.30 &   0.26 &            0.49 &   0.24 &         ...  &   ...  &     s &   Gr98 & (5)                                  							       			     \\
  HE 0206-1916      &          1.97 &   0.16 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   As05 & (4)                                  							       			     \\
  HE 0212-0557      &          2.25 &   0.06 &         2.27 &   0.22 &     2.21 &   ...  &            ...  &   ...  &         ...  &   ...  &    rs &   An89 & (2)                                  							       			     \\
  HE 0231-4016      &          1.47 &   0.23 &         1.22 &   0.28 &     1.53 &   0.27 &            ...  &   ...  &         ...  &   ...  &     s &   Gr98 & (5)                                  							       			     \\
  HE 0336+0113      &          2.12 &   ...  &         2.07 &   0.10 &     1.87 &   0.19 &            1.47 &   0.20 &         2.82 &   0.20 &    rs &   An89 & (6)                       									       			     \\
                    &          2.70 &   0.32 &         1.92 &   0.14 &     2.37 &   0.18 &            1.25 &   0.13 &    $<$2.45 &   ...  &    rs &   An89 & (2)                                  							       			     \\
  HE 0338-3945      &          2.41 &   0.07 &         2.28 &   0.16 &     2.16 &   0.18 &            1.94 &   0.17 &         3.10 &   0.11 &    rs &   As05 & (9)                                  							       			     \\
  HE 0400-2030      &          1.64 &   0.16 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   As05 & (4)                                  							       			     \\
  HE 0430-4404      &          1.62 &   0.23 &         1.41 &   0.26 &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   Gr98 & (5)                                  							       			     \\
  HE 0441-0652      &          1.11 &   0.24 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   As05 & (4)                                  							       			     \\
  HE 0507-1653      &          1.89 &   0.14 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   As05 & (4)                                  							       			     \\
  HE 0557-4840      &     $<$0.03 &   ...  &         ...  &   ...  &     ...  &   ...  &       $<$2.04 &   ...  &         ...  &   ...  &    no &   As05 & (10)                                  							       			     \\
 HE 1001-0243            &   0.41 &   0.20 &       0.55 &   0.20 &       0.80 &   0.10 &      --0.04 &   0.20 &   $<$1.38 &   ...& low-s & As05 & (52) \\
  HE 1005-1439      &          1.06 &   0.22 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   As05 & (4)                                  							       			     \\
  HE 1012-1540      &         --0.22 &   0.02 &         ...  &   ...  &     ...  &   ...  &       $<$1.92 &   ...  &    $<$3.10 &   ...  &    no &   An89 & (11)                                  							       			     \\
  HE 1031-0020      &          1.28 &   0.24 &         1.15 &   0.15 &     1.47 &   ...  &       $<$0.94 &   ...  &         2.83 &   ...  &     s &   An89 & (2)                                  							       			     \\
  HE 1105+0027      &          2.45 &   0.23 &         2.10 &   0.24 &     ...  &   ...  &            1.81 &   0.24 &         ...  &   ...  &    rs &   Gr98 & (5)                                  							       			     \\
  HE 1135+0139      &          1.13 &   0.24 &         0.93 &   0.25 &     1.17 &   0.27 &            0.33 &   0.25 &         ...  &   ...  &     s &   Gr98 & (5)                                  							       			     \\
  HE 1135-0344      &          ...  &   ...  &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr98 & (5)                                  							       			     \\
  HE 1150-0428      &         --0.54 &   0.14 &    $<$1.15 &   ...  &     ...  &   ...  &       $<$1.52 &   ...  &         ...  &   ...  &    no &   An89 & (2)                                  							       			     \\
  HE 1152-0355      &          1.58 &   ...  &         1.57 &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   As05 & (12)                                 							       			     \\
  HE 1157-0518      &          2.14 &   0.24 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    rs &   As05 & (4)                                  							       			     \\
  HE 1221-1948      &          ...  &   ...  &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr98 & (5)                                  							       			     \\
  HE 1249-3121      &          ...  &   ...  &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr98 & (5)                                  							       			     \\
  HE 1300+0157      &     $<$--0.85 &   ...  &         ...  &   ...  &     ...  &   ...  &       $<$1.56 &   ...  &    $<$2.78 &   ...  &    no &   As05 & (13)                                  							       			     \\
                    &          ...  &   ...  &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr98 & (5)                                  							       			     \\
                    &     $<$--0.56 &   ...  &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   An89 & (11)                                  							       			     \\
  HE 1300-0641      &         --0.77 &   0.26 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr98 & (5)                                  							       			     \\
  HE 1300-2201      &         --0.04 &   0.27 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr98 & (5)                                  							       			     \\
  HE 1305+0007      &          2.32 &   ...  &         2.56 &   ...  &     2.53 &   ...  &            1.97 &   ...  &         2.37 &   ...  &    rs &   As05 & (12)                                  							       			     \\
  HE 1305-0331      &          ...  &   ...  &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr98 & (5)                                  							       			     \\
  HE 1319-1935      &          1.89 &   0.26 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   As05 & (4)                                  							       			     \\
  HE 1327-2326      &     $<$1.58 &   0.12 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   As05 & (14,15,16)               					       			     \\
  HE 1327-2326      &     $<$1.58 &   0.12 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   As05 & (14,15,16)               					       			     \\
  HE 1330-0354      &         --0.47 &   0.26 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr98 & (5)                                  							       			     \\
  HE 1351-1049      &          0.13 &   0.25 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr98 & (5)                                  							       			     \\
  HE 1410-0004      &          1.13 &   0.24 &         ...  &   ...  &     ...  &   ...  &       $<$2.47 &   ...  &    $<$3.34 &   ...  &     s &   An89 & (2)                                  							       			     \\
 HE 1410+0213            &   0.05 &   0.20 &      --0.41 &   0.20 &       0.14 &   0.30 &     $<$0.60 &   ... &     $<$1.52 &   ...  & no & As05 & (52) \\
                    &          0.14 &   ...  &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   An89 & (2)                                  							       			     \\
  HE 1413-1954      &          ...  &   ...  &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr98 & (5)                                  							       			     \\
 HE 1419-1324            &   0.88 &   0.10 &       0.82 &   0.10 &       0.83 &   0.10 &       0.53 &   0.10 &       2.15 &   0.10 & low-s & As05 & (52) \\
  HE 1429-0551      &          1.57 &   0.17 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   As05 & (4)                                  							       			     \\
  HE 1430-1123      &          1.82 &   0.23 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   Gr98 & (5)                                  							       			     \\
  HE 1434-1442      &          1.30 &   0.17 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         2.35 &   ...  &     s &   An89 & (2)                                  							       			     \\
  HE 1443+0113      &          1.47 &   0.04 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   An89 & (2)                                  							       			     \\
  HE 1447+0102      &          2.70 &   0.14 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    rs &   As05 & (4)                                  							       			     \\
  HE 1509-0806      &          2.00 &   0.18 &         1.66 &   0.16 &     1.96 &   0.38 &       $<$1.00 &   ...  &         2.78 &   ...  &     s &   An89 & (2)                                  							       			     \\
  HE 1523-1155      &          1.72 &   0.14 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   As05 & (4)                                  							       			     \\
  HE 1528-0409      &          2.30 &   0.14 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    rs &   As05 & (4)                                  							       			     \\
  HE 2148-1247      &          2.41 &   0.15 &         2.30 &   0.20 &     2.30 &   0.20 &            2.03 &   0.10 &         3.22 &   0.20 &    rs &   Gr98 & (17)                                  							       			     \\
  HE 2150-0825      &          1.70 &   0.22 &         1.41 &   0.25 &     1.48 &   0.28 &            ...  &   ...  &         ...  &   ...  &     s &   Gr98 & (5)                                  							       			     \\
  HE 2158-0348      &          1.60 &   ...  &         1.87 &   0.10 &     2.46 &   0.41 &            1.37 &   0.20 &         3.42 &   0.20 &    rs &   An89 & (6)                       									       			     \\
                    &          1.66 &   0.15 &         1.54 &   0.21 &     1.96 &   0.21 &            0.87 &   0.15 &         2.77 &   ...  &     s &   An89 & (2)                                  							       			     \\
  HE 2221-0453      &          1.75 &   0.24 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   As05 & (4)                                  							       			     \\
  HE 2227-4044      &          1.38 &   0.24 &         1.28 &   0.26 &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   Gr98 & (5)                                  							       			     \\
  HE 2228-0706      &          2.50 &   0.16 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    rs &   As05 & (4)                                  							       			     \\
  HE 2232-0603      &          1.57 &   ...  &         1.22 &   0.05 &     0.83 &   0.07 &            0.67 &   0.10 &         2.12 &   0.20 &     s &   An89 & (6)                       									       			     \\
                    &          1.48 &   0.19 &         1.22 &   0.27 &     1.52 &   ...  &            ...  &   ...  &         1.72 &   ...  &     s &   An89 & (2)                                  							       			     \\
  HE 2240-0412      &          1.37 &   0.23 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   Gr98 & (5)                                  							       			     \\
  HE 2330-0555      &          1.22 &   0.25 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &  As05 & (4)                                  							       			     \\
  CS 22183-015      &          2.09 &   0.21 &         1.59 &   0.09 &     1.55 &   0.09 &            1.39 &   0.11 &         3.17 &   0.07 &    rs &   Gr96 & (18)                                  							       			     \\
 or HE  0058-0244   &          1.89 &   0.12 &         1.53 &   0.11 &     ...  &   ...  &            1.37 &   0.16 &         3.00 &   0.07 &    rs &   Gr98 & (19)                       									       			     \\
                    &          2.04 &   0.16 &         1.70 &   0.09 &     1.88 &   0.22 &            1.70 &   0.14 &         2.89 &   ...  &    rs &   An89 & (2)                                  							       			     \\
                    &          1.74 &   ...  &         1.87 &   0.10 &     2.01 &   ...  &            1.57 &   0.20 &         3.42 &   0.20 &    rs &   An89 & (6)        											       			     \\
  CS 22877-001      &         --0.49 &   0.17 &         ...  &   ...  &     ...  &   ...  &       $<$0.60 &   ...  &         ...  &   ...  &    no &   Gr96 & (20)                                  							       			     \\
  CS 22880-074      &          1.31 &   0.16 &         1.07 &   0.18 &     1.22 &   0.14 &            0.50 &   0.17 &         1.90 &   0.19 &     s &   Gr96 & (21,22)                       						       			     \\
                    &          1.34 &   ...  &         1.24 &   ...  &     ...  &   ...  &            0.55 &   ...  &         ...  &   ...  &     s &   An89 & (23)                                  							       			     \\
  CS 22881-036      &          1.93 &   ...  &         1.59 &   ...  &     ...  &   ...  &            1.00 &   ...  &         ...  &   ...  &     s &   An89 & (23)                                  							       			     \\
  CS 22887-048      &          2.00 &   0.14 &         1.73 &   0.08 &     ...  &   ...  &            1.49 &   0.14 &         3.40 &   0.07 &    rs &   Gr98 & (19)                       									       			     \\
 CS 22891-171            &   2.48 &   0.20 &       2.12 &   0.05 &       2.06 &   0.18 &       1.73 &   0.10 &       1.85 &   0.20 & rs & As05 & (52) \\
  CS 22892-052      &          0.96 &   0.05 &         1.09 &   0.05 &     1.03 &   0.07 &            1.66 &   0.05 &         1.35 &   0.15 &   rII &   Gr98 & (24)                                  							       			     \\
                    &          1.01 &   0.07 &         1.11 &   0.06 &     1.02 &   0.12 &            1.49 &   0.06 &         ...  &   ...  &   r   &   Gr98 & (25,26,27,29)            				       			     \\
                    &          1.19 &   0.23 &         1.02 &   0.24 &     ...  &   ...  &            1.54 &   0.24 &         ...  &   ...  &   r   &   Gr98 & (5)                                  							       			     \\
  CS 22898-027      &          2.23 &   0.14 &         2.13 &   0.18 &     2.13 &   0.13 &            1.88 &   0.15 &         2.84 &   0.19 &    rs &   Gr96 & (21,22)                       						       			     \\
                    &          2.26 &   0.16 &         2.19 &   0.14 &     ...  &   ...  &            1.91 &   0.11 &         2.89 &   0.07 &    rs &   Gr98 & (19)                       									       			     \\
                    &          2.27 &   ...  &         2.28 &   ...  &     ...  &   ...  &            1.94 &   ...  &         ...  &   ...  &    rs &   An89 & (23)                                  							       			     \\
 CS 22942-019            &   1.76 &   0.10 &       1.50 &   0.10 &       1.75 &   0.25 &       0.66 &   0.05 &   $<$2.03 &   ... & s & As05 & (52) \\
                    &          1.92 &   0.19 &         1.20 &   0.26 &     1.54 &   0.12 &            0.79 &   0.12 &    $<$1.60 &   ...  &     s &   Gr96 & (21,22)                       						       			     \\
                    &          1.50 &   ...  &         1.85 &   ...  &     ...  &   ...  &            0.80 &   ...  &         ...  &   ...  &     s &   An89 & (23)                                  							       			     \\
 CS 22945-017            &   0.55 &   0.20 &       0.99 &   0.20 &     $<$1.24 &  ... &    $<$1.20 &   ... &    $<$2.22 &   ... & no & As05 & (52) \\
  CS 22948-027      &          2.26 &   ...  &         2.32 &   ...  &     2.20 &   ...  &            1.88 &   ...  &         2.72 &   ...  &    rs &   Gr98 & (28)                                  							       			     \\
                    &          1.67 &   ...  &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    rs &   An89 & (23)                                  							       			     \\
                    &          1.85 &   0.14 &         1.91 &   0.15 &     1.99 &   0.14 &            1.57 &   ...  &         ...  &   ...  &    rs &   Gr96 & (20)                                  							       			     \\
                    &          2.31 &   0.15 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   As05 & (4)                                  							       			     \\
  CS 22949-037      &         --0.58 &   0.10 &         ...  &   ...  &     ...  &   ...  &       $<$0.04 &   ...  &         ...  &   ...  &    no &   Gr98 & (30)                                  							       			     \\
 or HE  2323-0256   &         --0.59 &   0.23 &         ...  &   ...  &     ...  &   ...  &       $<$0.80 &   ...  &         ...  &   ...  &    no &   An89 & (11)             									       				     \\
 CS 22956-028            &   0.16 &   0.20 &     $<$0.50 &   ...  &       0.85 &   0.30 &    $<$0.91 &  ...  &   $<$1.33 &  ... & no & As05 & (52) \\
                    &          0.56 &   0.15 &         0.67 &   0.20 &     ...  &   ...  &       $<$0.77 &   0.10 &         1.92 &   0.20 &    no &   An89 &  (6)                       									       			     \\
                    &          0.37 &   0.26 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   An89 & (31)                                  							       			     \\
  CS 22957-027      &         --0.95 &   0.12 &         ...  &   ...  &     ...  &   ...  &       $<$0.85 &   0.21 &         ...  &   ...  &    no &   An89 & (32)                                  							       			     \\
 or HE  2356-0410   &         --0.93 &   0.14 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   An89 & (33)                                  							       			     \\
                    &         --2.00 &   ...  &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   An89 & (23)                                  							       			     \\
                    &         --1.23 &   0.21 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr96 & (34)                                  							       			     \\
                    &         --0.71 &   0.25 &    $<$0.59 &   ...  &     ...  &   ...  &       $<$1.04 &   ...  &         ...  &   ...  &    no &   An89 & (2)                                  							       			     \\
                    &         --0.91 &   ...  &         0.47 &   0.20 &     ...  &   ...  &           --0.53 &   0.20 &    $<$1.42 &   0.20 &    r  &   An89 &  (6)      											       			     \\
  CS 22958-042      &     $<$--0.53 &   0.16 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   As05 & (35)                                  							       			     \\
  CS 22960-053      &          0.86 &   0.14 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   As05 & (4)                                  							       			     \\
  CS 22967-007      &          2.09 &   0.12 &         1.47 &   0.10 &     1.69 &   0.05 &            0.77 &   0.20 &         2.82 &   0.10 &     s &   An89 &  (6)                       									       			     \\
  CS 29495-042      &          1.83 &   0.18 &         1.27 &   0.20 &     1.48 &   0.05 &            0.77 &   0.20 &         1.32 &   0.20 &     s &   An89 &  (6)                       									       			     \\
  CS 29497-030      &          2.17 &   0.15 &         2.10 &   0.10 &     2.14 &   0.17 &            1.44 &   0.15 &         3.55 &   ...  &    rs &   Gr98 & (36)                                  							       			     \\
                    &          2.32 &   0.11 &         2.22 &   0.10 &     2.10 &   0.10 &            1.99 &   0.10 &         3.65 &   0.13 &    rs &   An89 & (37)                                  							       			     \\
  CS 29497-034      &          2.03 &   ...  &         2.12 &   ...  &     1.95 &   ...  &            1.80 &   ...  &         2.95 &   ...  &    rs &   Gr98 & (28)                                  							       			     \\
                    &          1.79 &   0.10 &         1.87 &   0.20 &     ...  &   ...  &       $<$1.57 &   0.10 &         4.52 &   0.20 &     s &   An89 &  (6)                       									       			     \\
 or HE 0039-2635      &          2.23 &   0.13 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    rs &   As05 & (4)                                  							       			     \\
  CS 29498-043      &         --0.46 &   0.20 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr96 & (38)                                  							       			     \\
                    &         --0.45 &   0.20 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   Gr96 & (34)                                  							       			     \\
  CS 29502-092      &         --0.82 &   0.27 &         ...  &   ...  &     ...  &   ...  &       $<$0.40 &   ...  &         ...  &   ...  &    no &   Gr96 & (20)                                  							       			     \\
                    &         --1.37 &   0.07 &    $<$0.06 &   ...  &     ...  &   ...  &       $<$0.21 &   ...  &    $<$1.53 &   ...  &    no &   Gr98 & (19)                       									       			     \\
  CS 29503-010      &          1.50 &   0.16 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   As05 & (4)                                  							       			     \\
  CS 29509-027      &          1.33 &   0.13 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   An89 & (31)                                  							       			     \\
  CS 29526-110      &          2.11 &   0.17 &         1.69 &   0.24 &     2.01 &   0.22 &            1.73 &   0.20 &         3.30 &   0.24 &    rs &   Gr96 & (21,22)                       						       			     \\
  CS 29528-028      &          3.27 &   0.20 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    rs &   As05 & (4)                                  							       			     \\
  CS 29528-041      &          0.97 &   0.10 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   As05 & (35)                                  							       			     \\
  CS 30301-015      &          1.45 &   0.16 &         0.84 &   0.25 &     1.16 &   0.15 &            0.20 &   0.18 &         1.70 &   0.24 &     s &   Gr96 & (21,22)                       						       			     \\
  CS 30314-067      &         --0.57 &   0.14 &         ...  &   ...  &     ...  &   ...  &       $<$--0.50 &   ...  &         ...  &   ...  &   no  & Gr96 & (20)                                  							       			     \\
  CS 30315-091      &          1.62 &   0.15 &         0.87 &   0.20 &     1.41 &   0.12 &            0.37 &   0.20 &         1.92 &   0.30 &     s &   An89 &  (6)                       									       			     \\
 CS 30322-023            &   0.52 &   0.05 &       0.46 &   0.05 &       0.59 &   0.24 &      --0.63 &   0.10 &       1.49 &   0.20 & low-s  & As05 & (52) \\
                    &          0.59 &   0.17 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   As05 & (4)                                  							       			     \\
  CS 30323-107      &          1.90 &   0.17 &         1.12 &   0.10 &     1.46 &   0.06 &       $<$0.57 &   0.10 &         2.52 &   0.20 &     s &   An89 &  (6)                       									       			     \\
  CS 30338-089      &          2.22 &   0.15 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    rs &   As05 & (4)                                  							       			     \\
  CS 30338-089      &          1.87 &   0.38 &         1.57 &   0.20 &     3.09 &   0.26 &            1.77 &   0.20 &         3.72 &   0.20 &    rs &   An89 &  (6)                       									       			     \\
  CS 31062-050      &          2.30 &   0.15 &         2.44 &   0.20 &     2.10 &   0.12 &            1.84 &   0.13 &         2.90 &   0.24 &    rs &   Gr96 & (21,22)                       						       			     \\
                    &          2.80 &   0.20 &         2.12 &   0.12 &     2.02 &   0.16 &            1.79 &   0.07 &         2.81 &   0.15 &    rs &   Gr96 & (39)                                  							       			     \\
  CS 31080-095      &          0.77 &   0.15 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   As05 & (35)                                  							       			     \\
  BS 16080-175      &          1.55 &   0.07 &         1.65 &   0.07 &     ...  &   ...  &            1.05 &   0.07 &         2.60 &   0.07 &    rs &   Gr98 & (19)                       									       			     \\
  BS 16929-005      &         --0.48 &   0.12 &         ...  &   ...  &     ...  &   ...  &            ...  &   ...  &         ...  &   ...  &    no &   As05 & (4)                                  							       			     \\
  BS 17436-058      &          1.61 &   0.14 &         1.49 &   0.17 &     ...  &   ...  &            1.17 &   0.11 &         2.26 &   0.07 &    rs &   Gr98 & (19)                       									       			     \\
 HD 26                   &   1.85 &   0.20 &       1.39 &   0.10 &       1.67 &   0.10 &       0.75 &   0.10 &       2.02 &   0.10 & s & As05 & (52) \\
  HD 5223           &          1.82 &   ...  &         1.76 &   ...  &     1.87 &   ...  &            ...  &   ...  &    $<$2.21 &   ...  &     s &   As05 & (12)                                  							       			     \\
HD 187861               &   1.39 &   0.10 &       1.73 &   0.20 &       1.37 &   0.14 &       1.34 &   0.20 &       2.86 &   0.20 & rs & As05 & (52) \\
 HD 196944               &   1.22 &   0.20 &       0.86 &   0.05 &       0.99 &   0.12 &       0.22 &   0.20 &       1.99 &   0.10 & s & As05 & (52) \\
                    &          1.10 &   0.19 &         0.91 &   0.28 &     1.01 &   0.19 &            0.17 &   0.19 &         1.90 &   0.24 &     s &   Gr96 & (21,22)                       						       			     \\
                    &          1.26 &   0.30 &         ...  &   ...  &     0.61 &   ...  &            ...  &   ...  &         ...  &   ...  &     s &   An89 & (40)                                  							       			     \\
 HD 206983               &   0.92 &   0.20 &       0.86 &   0.05 &       1.06 &   0.10 &       0.57 &   0.10 &       1.49 &   0.20 & s  & As05 & (52) \\
 HD 224959               &   2.19 &   0.20 &       2.03 &   0.05 &       1.91 &   0.09 &       1.74 &   0.10 &       3.06 &   0.10 & rs & As05 & (52) \\
HKII17435-00532    &          0.86 &   0.29 &         0.78 &   0.21 &     1.17 &   0.20 &            0.48 &   0.20 &         ...  &   ...  & low-s &   Gr98 & (51)                                  \\
  LP 625-44         &          2.74 &   0.20 &         2.46 &   0.13 &     2.27 &   0.12 &            1.97 &   0.20 &         2.55 &   0.20 &    rs &   Gr96 & (41)                                  							       			     \\
                    &          2.81 &   0.24 &         2.40 &   0.26 &     2.22 &   0.23 &            1.72 &   0.24 &         2.60 &   0.22 &    rs &   Gr96 & (42)                                  							       			     \\
  LP 706-7          &          2.01 &   0.14 &         1.81 &   0.19 &     1.86 &   0.31 &            1.40 &   0.20 &         2.28 &   0.20 &    rs &   Gr96 & (41)                                  							       			     \\
 or CS 31062-012      &          1.98 &   0.16 &         2.02 &   0.18 &     2.12 &   0.13 &            1.62 &   0.14 &         2.40 &   0.19 &    rs &   Gr96 & (21,22)                       						       			     \\
   G 77-61          &     $<$1.00 &   ...  &         ...  &   ...  &     ...  &   ...  &       $<$3.00 &   ...  &         ...  &   ...  &    no &   Gr98 & (43)                                  							       			     \\
   V 543Oph         &          0.34 &   ...  &         0.37 &   0.09 &     ...  &   ...  &            0.65 &   0.20 &         ...  &   ...  &    no &   Gr98 & (44)    \\
\hline	
\end{tabular}
\end{table}